Author's response to reviews

Title: Psychological and social consequences among mothers suffering from perinatal loss: perspective from a low income country.

Authors:

Kaniz Gausia (k.gausia@ecu.edu.au)
Allisyn C Moran (AMoran@savechildren.org)
Mohammed Ali (m.ali@curtin.edu.au)
David Ryder (david.ryder@ecu.edu.au)
Colleen Fisher (colleen.fisher@uwa.edu.au)
Marge Koblinsky (marge_koblinsky@jsi.com)

Version: 3 Date: 19 May 2011

Author's response to reviews: see over
May 18, 2011

To
Editor in Chief, BMC Public Health
Biomed Central Ltd
Floor 6, 236 Gray’s Inn Road, London
WC1X 8HB, United Kingdom

Dear Dr. Norton,

Thank you for giving us the opportunity to revise our manuscript “Psychological and social consequences among mothers suffering from perinatal loss: perspective from a low income country’. We would also like to convey our sincere thanks to the reviewers for their valuable expert opinions in improving the quality of the manuscript. We have addressed the reviewers’ comments and made the necessary changes to the manuscript as suggested by them. Our responses to each of the comments by the reviewers are attached.

Thank you.

Kaniz Gausia
Postdoctoral Research Fellow
Public Health
School of Exercise, Biomedical & Health Sciences
Edith Cowan University
270 Joondalup Drive
JOONDALUP WA 6027
AUSTRALIA
Phone:  61 8 6304 2380
Fax:    61 8 6304 5449
Email:  k.gausia@ecu.edu.au